Author's response to reviews

Title: Moderate Alcohol Consumption is Associated with Better Endothelial Function: a cross-sectional study.

Authors:

Keiko Suzuki (majokko7174@gmail.com)
Mitchell Elkind (MSE13@columbia.edu)
Bernadette Boden-Albala (bb87@columbia.edu)
Zhezhen Jin (zi7@columbia.edu)
Grace Berry (gb2221@columbia.edu)
Marco Di Tullio (md42@columbia.edu)
Ralph Sacco (RSacco@med.miami.edu)
Shunichi Homma (sh23@columbia.edu)

Version: 4 Date: 6 February 2009

Author's response to reviews: see over
Feb 5 2009

Dear Editor-in-Chief
Dr. Melissa Norton

Enclosed please find our revised manuscript entitled “Moderate alcohol consumption is associated with better endothelial function.” for possible publication in *BMC cardiovascular disorder* as a research article.
This includes a text of the manuscript and 2 tables in one file.

The name, address, telephone number, and FAX number of the corresponding author are as follows:

Name: Shunichi Homma, M.D.
Address: Division of Cardiology, Department of Medicine,
         Columbia University College of Physicians and Surgeons,
         630 West 168th Street, PH3-342, New York, NY,
         United States 10032
Telephone number: 1-212-305-3068
FAX number: 1-212-305-5280
Email address sh23@columbia.edu

In this study, we evaluated the association between alcohol consumption and flow-mediated dilation of the brachial artery assessed by high-resolution ultrasonography. We found moderate alcohol consumption was independently associated with better flow-mediated dilation. I declare that neither the paper submitted nor any similar paper has been or will be submitted to or published in any other primary journal. All of the authors are aware of and agree to the content of the paper and to being listed as an author on the paper.

In this revised version, we added acknowledgements section and corrected the format of tables as your direction.

I sincerely hope that you will give favorable consideration to our article and find this manuscript acceptable for publication in *BMC cardiovascular disorders*.

Sincerely yours,

Shunichi Homma, MD
Division of Cardiology, Department of Medicine,
Columbia University College of Physicians and Surgeons